Title: Metabolic and Endocrinological complications in Beta-thalassemia major: A Multicenter Study in Tehran

Authors:

Dr Alireza Abdollah Shamshirsaz (alirezashamshirsaz@yahoo.com)
Mir Reza Bekheirnia (rezabkh@hotmail.com)
Dr Mohammad Kamgar (m_kamgar@yahoo.com)
Dr Nima Pourzahedgilani (nimapg@yahoo.com)
Navid Bouzari (nbouzari@yahoo.com)
Mohammad Reza Habibzadeh (mrezahabibzadeh@yahoo.com)
Seyed Reza Hashemi (r_hashemi@hotmail.com)
Amir Hooshang Abdollah Shamshirsaz (shamshir_a1978@yahoo.com)
Shahriar Aghakhani (aghakhani@parsonline.net)
Dr Hooman Homayoun (alirezashamshirsaz@yahoo.com)
Dr Bagher Larijani (emrc@sina.tums.ac.ir)

Version: 4 Date: 3 Aug 2003

We would be pleased if you could consider our titles (Dr) for all the authors. (eg. Dr Mir Reza Bekheirnia or Dr Mohammad Kamgar ...)

In Title: "endocrinologic" has changed to "endocrinological"
In Methods, first paragraph, patients selection has changed as follows:

Thalassemic patients in Tehran are all registered in 12 clinics in order to receive treatment. These clinics were sorted in a list according to the number of registered patients. From every two consecutive centers, one was selected with simple random method in order to select six thalassemia clinics. Number of patients to be selected from each clinic was calculated with proportional to size method based on registered cases in each of these clinics. In each clinic, patients were selected with simple random method by using computerized tables (EPI INF). Two hundred fifty eight adolescent homozygous beta-thalassemia patients were recruited, among them, 220 (51.5% male) accepted to enter the study. Table 1 show...